\documentclass[12pt,reqno]{article}

\usepackage[usenames]{color}
\usepackage{amssymb}
\usepackage{amsmath}
\usepackage{amsthm}
\usepackage{amsfonts}
\usepackage{amscd}
\usepackage{graphicx}

\usepackage[colorlinks=true,
linkcolor=webgreen,
filecolor=webbrown,
citecolor=webgreen]{hyperref}

\definecolor{webgreen}{rgb}{0,.5,0}
\definecolor{webbrown}{rgb}{.6,0,0}

\usepackage{color}
\usepackage{fullpage}
\usepackage{float}

\usepackage{psfig}
\usepackage{graphics}
\usepackage{latexsym}
\usepackage{epsf}

\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{.1in}
\setlength{\evensidemargin}{.1in}
\setlength{\topmargin}{-.1in}
\setlength{\textheight}{8.4in}

\newcommand{\seqnum}[1]{\href{https://oeis.org/#1}{\rm \underline{#1}}}


\begin{document}
\begin{thebibliography}{99}

\bibitem{Bitim} B.\ D.\ Bitim and N.\ Topal, $(p,q)$-Fibonacci and $(p,q)$-Lucas sums by the derivatives of some polynomials, \textit{Commun. Fac. Sci. Univ. Ank. Ser A1 Math. Stat.} \textbf {68} (2019), 1733--1741.

\bibitem{Bramham} A.\ Bramham and M.\ Griffiths, Combinatorial interpretation of some convolution identities, \textit{Fibonacci Quart.} \textbf{54} (2016), 335--339.

\bibitem{Camp} H.\ Campos, P.\ Catarino, A.\ P.\ Aires, P.\ Vasco, and A.\ Borges, On some identities of $k$-Jacobsthal-Lucas numbers, \textit{Int. J. Math. Anal.} \textbf {8} (2014), 489--494.

\bibitem{Falcon11} S.\ Falc\'{o}n, On the $k$-Lucas numbers, \textit{Int. J. Contemp. Math. Sci.} \textbf {6} (2011), 1039--1050.

\bibitem{Falcon2012} S.\ Falc\'{o}n, On the Lucas triangle and its relationship with the $k$-Lucas numbers, \textit{J. Math. Comput. Sci.} \textbf {2} (2012), 425--434.

\bibitem{Fal32} S.\ Falc\'{o}n and \'{A}.\ Plaza, On the Fibonacci $k$-numbers, \textit{Chaos Solitons Fract.} \textbf{32} (2007), 1615--1624.

\bibitem{Fal33} S.\ Falc\'{o}n and \'{A}.\ Plaza, The $k$-Fibonacci sequence and the Pascal 2-triangle, \textit{Chaos Solitons Fract.} \textbf{33} (2007), 38--49.

\bibitem{Griffiths} M.\ Griffiths and A.\ Bramham, The Jacobsthal numbers: two results and two questions, \textit{Fibonacci Quart.} \textbf{53} (2015), 147--151.

\bibitem{Haukkanen} P.\ Haukkanen, A note on specially multiplicative arithmetic functions, \textit{Fibonacci Quart.} \textbf{26} (1988), 325--327.

\bibitem{HoroFQ} A.\ F.\ Horadam, Basic properties of a certain generalized sequence of numbers, \textit{Fibonacci Quart.} \textbf{3} (1965), 161--176.

\bibitem{Horad} A.\ F.\ Horadam, Jacobsthal and Pell curves, \textit{Fibonacci Quart.} \textbf{26} (1988), 77--83.

\bibitem{Jhala} D.\ Jhala, K.\ Sisodyiya, and G.\ P.\ S.\ Rathore, On some identities for $k$-Jacobsthal numbers, \textit{Int. J. Math. Anal.} \textbf {7} (2013), 551--556.

\bibitem{Kalm} D.\ Kalman and R.\ Mena, The Fibonacci numbers---exposed, \textit{Math. Mag.} \textbf {76} (2003), 167--181.

\bibitem{Kesava} P.\ Kesava Menon, A class of quasi-fields having isomorphic additive and multiplicative groups, \textit{J. Ind. Math. Soc.} \textbf{27} (1963), 71--90.

\bibitem{Koshy} T.\ Koshy and M.\ Griffiths, Some Gibonacci convolutions with dividends, \textit{Fibonacci Quart.} \textbf{56} (2018), 237--245.

\bibitem{McC} P.\ J.\ McCarthy, \textit{Introduction to Arithmetical Functions}, Springer-Verlag, 1986.

\bibitem{McCarthy} P.\ J.\ McCarthy and R.\ Sivaramakrishnan, Generalized Fibonacci sequences via arithmetical functions, \textit{Fibonacci Quart.} \textbf{28} (1990), 363--370.

\bibitem{Panwar} Y.\ K.\ Panwar, A.\ Mansuri, and J.\ Bhandari, On the sums of $k$-Lucas numbers, \textit{Adv. Appl. Math. Sci.} \textbf{20} (2021), 3427--3434.

 \bibitem{Schk} E.\ D.\ Schwab and G.\ Schwab, $k$-Fibonacci numbers and M\"{o}bius functions, \textit{Integers} \textbf{22} (2022), \#A64.

\bibitem{Sch94} E.\ D.\ Schwab and G.\ Schwab, Dirichlet convolution and the Binet formula, \textit{Integers} \textbf{23} (2023), \#A94.

\bibitem{SchMB} E.\ D.\ Schwab and G.\ Schwab, A note on $(a,b)$-Fibonacci sequences and specially multiplicative arithmetic functions, \textit{Math. Bohem.} \textbf{149} (2024), 237--246.

\bibitem{SchAMM} E.\ D.\ Schwab and G.\ Schwab, Jacobsthal analogues of two Fibonacci-Lucas identities and a generalization, \textit{Elem. Math.}, to appear.

\bibitem{Siva} R.\ Sivaramakrishnan, \textit{Classical Theory of Arithmetic Functions}, Marcel Dekker, 1989.

\bibitem{Szakacs} T.\ Szak\'{a}cs, Convolution of second order linear recursive sequences II, \textit{Commun. Math.} \textbf{25} (2017), 137--148.

\bibitem{Uygun} S.\ Uygun and H.~\ Eldogan, Properties of $k$-Jacobsthal and $k$-Jacobsthal Lucas sequences, \textit{Gen. Math. Notes} \textbf {36} (2016), 34--47.

\end{thebibliography}


\begin{center}
\epsfxsize=4in
\leavevmode\epsffile{logo129.eps}
\end{center}


\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}


\begin{center}
\vskip 1cm{\LARGE\bf Fibonacci Numbers via Dirichlet Convolution \\
\vskip .1in
and Lucas Numbers via Unitary Convolution
}
\vskip 1cm
\large
Emil Daniel Schwab \\
Department of Mathematical Sciences\\
The University of Texas at El Paso\\
El Paso, TX 79968 \\
USA \\
\href{mailto:eschwab@utep.edu }{\tt eschwab@utep.edu}\\
\end{center}

\vskip .2 in


\begin{abstract}
We apply the multiplicative arithmetic functions strategy to sequences
defined by linear two-term recurrences. We associate Fibonacci
numbers with Dirichlet convolution and Lucas numbers with unitary
convolution. Some Fibonacci-Lucas sums are shown as applications by
using the Kesava Menon quasi-distributive law (a distributive-like
property of Dirichlet convolution over unitary convolution) and
the Fibonacci-Lucas quotients under the Dirichlet and the unitary
convolutions.
\end{abstract}

\section{Introduction and preliminaries}
\label{sec1}

\subsection{Brief introduction and motivation}

Haukkanen \cite{Haukkanen}, McCarthy, and Sivaramakrishnan \cite{McCarthy} were the first to create a bridge between the two chapters of elementary number theory: arithmetic functions and Fibonacci numbers.
In two recent papers, E. D. Schwab and G. Schwab  \cite{Schk, SchMB} found new links between specially multiplicative (quadratic) arithmetic functions and generalized Fibonacci numbers. These two chapters of number theory have interacted, creating the possibility to widen the field of action for both. The basic identities of Fibonacci numbers, such as Cassini's, d'Ocagne's, Catalan's, Vajda's, and Honsberger's identities, are all particular cases of Busche-Ramanujan identities from the theory of specially multiplicative arithmetic functions (see \cite[Theorem 3.1]{SchMB}). This relationship leads to interesting paths in the parallel approach of these two chapters of elementary number theory.

This paper focuses on some Fibonacci-Lucas-type identities using the Dirichlet convolution and the unitary convolution of arithmetic functions. The Kesava Menon quasi-distributive law of Dirichlet convolution over unitary convolution 
(see Equation (\ref{eq.Distributive}) below) 
is a new source for Fibonacci-Lucas-type identities. For instance, the following Fibonacci-Lucas-type identities ($F_{i}$, $L_{i}$, $J_{i}$, and $j_{i}$ are the $i^{\rm th}$ Fibonacci, Lucas, Jacobsthal, and Jacobsthal-Lucas numbers, respectively):
\begin{align}
& \sum_{i=0}^{n}L_{n-i}J_{i}=j_{n+1}-L_{n+1}\ \quad\ (\text{Griffiths and Bramham}\ \cite[\text{Section}\ 2]{Griffiths}\ \label{eq.Griffiths}) \\
\text{and}\ \quad\ &\sum_{i=0}^{n}j_{n-i}F_{i}=j_{n+1}-L_{n+1}\ \quad\ (\text{Koshy and Griffiths}\ \cite[\text{Equation}\ (2.7)]{Koshy} \label{eq.Koshy})
\end{align}
are particular cases of a general identity (see Corollary \ref{Cor.}) which is proved below using the Kesava Menon quasi-distributive law.

Essentially,
this paper is not only about an alternative technique for proving identities; but also about the close relationship between arithmetic functions and Fibonacci numbers, and their effects on each other. Fibonacci numbers are associated with Dirichlet convolution and Lucas numbers with unitary convolution of two completely multiplicative functions (see Equation (\ref{eq.Di-uni}) and 
Theorem~\ref{Th.first} (i)). This simple observation motivated us to choose Fibonacci-Lucas relations for our applications.

The paper is self-contained and the next Subsections \ref{subsect:Fib-type} and \ref{subsect.:Arithm.F.} include all results and notation used in Sections \ref{sect:Fib-Luc} and \ref{sect.:Appl.}. Section \ref{sect:Fib-Luc} is devoted to Fibonacci $\mathcal{F}_{a,b}$ and Lucas $\mathcal{L}_{a,b}$ multiplicative arithmetic functions; in particular to Dirichlet and unitary quotients $\left[ \frac{\mathcal{F}_{a,b}}{\mathcal{L}_{a,b}} \right]_{D}$ and $\left[ \frac{\mathcal{L}_{a,b}}{\mathcal{F}_{a,b}} \right]_{U}$, respectively. Both quotients are used in Subsections \ref{Subs.3.1} and \ref{Subs.3.2} as a starting point to establish some Fibonacci-Lucas sums. Using the Kesava Menon quasi-distributive law, we prove Theorem \ref{Th.compl} and implicitly Corollary \ref{Cor.}.
Perhaps Subsection \ref{Subs.3.4} best illustrates how the strategy of approaching Fibonacci numbers works using the tools of multiplicative arithmetic functions.

\subsection{Fibonacci-type sequences and their companions} \label{subsect:Fib-type}

Throughout this paper, the focus of attention is on the Fibonacci and the Lucas sequences in their general form (i.e., $(a,b)$-Fibonacci and $(a,b)$-Lucas sequences). 

The companion sequence of the famous Fibonacci sequence $\{F_{m}\}_{m\geq0}$ (\seqnum{A000045}),
\begin{equation*}
F_{0}=0, F_{1}=1, \qquad F_{m+2}=F_{m+1}+F_{m},
\end{equation*}
is the Lucas sequence $\{L_{m}\}_{m\geq0}$ (\seqnum{A000032}) defined by the same recurrence relation but with different initial conditions
\begin{equation*}
L_{0}=2, L_{1}=1, \qquad L_{m+2}=L_{m+1}+L_{m}.
\end{equation*}
Some well-known Fibonacci-type numbers modify the recurrence relation slightly, leaving the initial conditions unchanged. The Pell numbers $\{P_{m}\}_{m\geq0}$ (\seqnum{A000129}), the Jacobsthal numbers $\{J_{m}\}_{m\geq0}$ (\seqnum{A0001045}), the $k$-Fibonacci numbers $\{F_{k,m}\}_{m\geq0}$ (\cite{Fal32,Fal33}), the $k$-Jacobsthal numbers $\{J_{k,m}\}_{m\geq0}$ (\cite{Jhala}),
are such numbers:
\begin{align*}
P_{0}=0,  P_{1}& =1, \qquad P_{m+2} =2P_{m+1}+P_{m}; \\
J_{0}=0,  J_{1}& =1, \qquad J_{m+2} =J_{m+1}+2J_{m}; \\
F_{k,0}=0,  F_{k,1}& =1, \qquad F_{k,m+2} =kF_{k,m+1}+F_{k,m}; \\
J_{k,0}=0,  J_{k,1}& =1, \qquad J_{k,m+2} =kJ_{k,m+1}+2J_{k,m}.
\end{align*}
The companion of the Pell sequence (say Pell-Lucas sequence) $\{Q_{m}\}_{m\geq0}$ (\cite[\seqnum{A002203}]{Horad}), the Jacobsthal-Lucas $\{j_{m}\}_{m\geq0}$ (\cite[\seqnum{A014551}]{Horad}), the $k$-Lucas $\{L_{k,m}\}_{m\geq0}$ (\cite{Falcon11,Falcon2012}), and the $k$-Jacobsthal-Lucas $\{j_{k,m}\}_{m\geq0}$ (\cite{Camp,Uygun}) sequences
are defined by
\begin{align*}
Q_{0}=2,Q_{1}& =2, \qquad Q_{m+2}=2Q_{m+1}+Q_{m}; \\
j_{0}=2,j_{1}& =1, \qquad j_{m+2}=j_{m+1}+2j_{m}; \\
L_{k,0}=2,L_{k,1}& =k, \qquad L_{k,m+2}=kL_{k,m+1}+L_{k,m}; \\
\end{align*}
and
\begin{equation*}
j_{k,0}=2,j_{k,1}=k, \qquad j_{k,m+2}=kj_{k,m+1}+2j_{k,m},
\end{equation*}
respectively.

In the mathematical literature, generalizations of Fibonacci numbers have an extensive place. For fixed real numbers $a$ and $b$ (can also be complex), let $\mathcal{R}(a,b)$ denote (see \cite{Kalm}) the set of all sequences $\{A_{m}\}_{m\geq0}$ with initial values $A_{0}$ and $A_{1}$ for which all succeeding terms are determined by
\begin{equation*}
A_{m+2}=aA_{m+1}+bA_{m}.
\end{equation*}
The set of sequences $\mathcal{R}(a,b)$ is a two-dimensional subspace of $\mathbb{R}^{\infty}$. Two distinguished elements of $\mathcal{R}(a,b)$ are the 
$(a,b)$-Fibonacci sequence, say $\{F_{m}^{a,b}\}_{m\geq0}$,
and the $(a,b)$-Lucas sequence, say $\{L_{m}^{a,b}\}_{m\geq0}$, defined by
\begin{equation*}
F_{0}^{a,b}=0,F_{1}^{a,b}=1, \qquad F_{m+2}^{a,b}=aF_{m+1}^{a,b}+bF_{m}^{a,b}
\end{equation*}
and
\begin{equation*}
L_{0}^{a,b}=2,L_{1}^{a,b}=a, \qquad L_{m+2}^{a,b}=aL_{m+1}^{a,b}+bL_{m}^{a,b},
\end{equation*}
respectively. The $(a,b)$-Fibonacci number Binet formula and the $(a,b)$-Lucas number Binet formula are (see \cite[Equations (9) and (10)]{Kalm})
\begin{equation} \label{eq.Binet}
F_{m}^{a,b}=\frac{\alpha^{m}-\beta^{m}}{\alpha-\beta} \qquad \text{and} \qquad L_{n}^{a,b}=\alpha^{m}+\beta^{m},
\end{equation}
respectively, where
\begin{equation*}
\alpha+\beta=a \qquad \text{and} \qquad \alpha\beta=-b.
\end{equation*}
The Lucas-Fibonacci basic connection is stated by the following equation (see \cite[Equation (5)]{Kalm}):
\begin{equation} \label{eq.connect}
L_{m+1}^{a,b}=F_{m+2}^{a,b}+bF_{m}^{a,b}.
\end{equation}

The sequence $\{H_{m}^{a,b}(r,s)\}_{m\geq0}$ (extensively studied by Horadam \cite{HoroFQ}) as an element of $\mathcal{R}(a,b)$ is a second-order linear recurrence sequence depending on four parameters (two initial values and two in the defining recursion itself):
\begin{equation*}
H_{0}^{a,b}(r,s)=r,H_{1}^{a,b}(r,s)=s \qquad \text{and} \qquad H_{m+2}^{a,b}(r,s)=aH_{m+1}^{a,b}(r,s)+bH_{m}^{a,b}(r,s).
\end{equation*}
Extension of the definition of $\{H_{m}^{a,b}(r,s)\}_{m\geq0}$ to negative subscript is provided by the recurrence relation
\begin{equation*}
H_{-n}^{a,b}(r,a)=\frac{1}{b}(H_{-n+2}^{a,b}+aH_{-n+1}^{a,b}).
\end{equation*}
Now, $F_{m}^{a,b}=H_{m}^{a,b}(0,1)$, $L_{m}^{a,b}=H_{m}^{a,b}(2,a),$ and $F_{-1}^{a,b}=\frac{1}{b},$ $L_{-1}^{a,b}=-\frac{a}{b},$ etc.

\subsection{Arithmetic functions} \label{subsect.:Arithm.F.}
For information on arithmetic functions, we refer to McCarthy's \cite{McC} and Sivaramakrishnan's \cite{Siva} books.

An arithmetic function is a complex-valued function defined on the set of positive integers. The Dirichlet convolution $f\ast{g}$ and the unitary convolution $f\sqcup{g}$ of two arithmetic functions $f$ and $g$ are defined by
 \begin{equation*}
(f\ast{g})(n)=\sum_{d\mid{n}}f(d)g\left(\frac{n}{d}\right) \qquad \text{and} \qquad (f\sqcup{g})(n)=\sum_{d'\mid\mid{n}}f(d')g\left(\frac{n}{d'}\right),
\end{equation*}
respectively, where the summation is over the positive divisors $d$ of $n$ in the first case, and over the unitary divisors $d'$ (i.e., $d'\mid{n}$ and $\gcd(d',\frac{n}{d'})=1$) of $n$ in the second case.

An arithmetic function $f$ is called multiplicative if $f(1)=1$ and $f(nn')=f(n)f(n')$ whenever $\gcd(n,n')=1$. A multiplicative arithmetic function $f$ is said to be completely multiplicative if $f(nn')=f(n)f(n')$ holds for all positive integers $n$ and $n'.$ The set of multiplicative arithmetic functions $\mathcal{M}$ forms a group $(\mathcal{M},\ast)$ under the Dirichlet convolution and simultaneously forms a group $(\mathcal{M},\sqcup)$ under the unitary convolution. These two groups are isomorphic abelian groups and the multiplicative arithmetic function $e$ defined by $e(n)=0$ if $n=1$ and $e(n)=0$ otherwise is the identity element in both groups. In fact, $(\mathcal{M},\sqcup,\ast)$ is a quasi-field in the sense of Kesava Menon \cite{Kesava} with the following quasi-distributive law:
\begin{equation} \label{eq.Distributive}
f\ast(g\sqcup{h})\sqcup{f}=(f\ast{g})\sqcup(f\ast{h}),
\end{equation}
that we call the Kesava Menon quasi-distributive law.

Note that a multiplicative arithmetic function is completely determined by its values at the prime powers.

If $f$ and $g$ are two multiplicative arithmetic functions, then the quotients $\left[ \frac{f}{g} \right]_{D}$ and $\left[ \frac{f}{g} \right]_{U}$ denote the multiplicative arithmetic functions for which $f=\left[ \frac{f}{g} \right]_{D}\ast{g}$ and $f=\left[\frac{f}{g} \right]_{U}\sqcup{g}$, respectively. In other words,
\begin{equation*}
\left[ \frac{f}{g} \right]_{D}=f\ast{g}^{-1} \qquad \text{and} \qquad \left[\frac{f}{g} \right]_{U}=f\sqcup{g}^{<},
\end{equation*}
where $g^{-1}$ and $g^{<}$ are the inverse elements of $g$ in the groups $(\mathcal{M},\ast)$ and $(\mathcal{M},\sqcup)$, respectively.

\section{Fibonacci-Lucas quotients under the Dirichlet and the unitary convolutions} \label{sect:Fib-Luc}
For any positive integer $n$, let $\Omega(n)$ be the number of prime factors of $n$ counted with multiplicity. If $c$ is a non-zero complex number then the arithmetic function $c^{\Omega(n)}$ (in short $c^{\Omega}$) is completely multiplicative.
\begin{definition}
Given two complex numbers $a$ and $b$, $b\neq0$, we define the $(a,b)$-Fibonacci $\mathcal{F}_{a,b}$, and the $(a,b)$-Lucas $\mathcal{L}_{a,b}$  multiplicative arithmetic functions as
\begin{equation} \label{eq.Di-uni}
\mathcal{F}_{a,b}=\alpha^{\Omega}\ast{\beta^{\Omega}} \qquad \text{and} \qquad \mathcal{L}_{a,b}=\alpha^{\Omega}\sqcup{\beta^{\Omega}},
\end{equation}
respectively, where
\begin{equation}\ \label{eq.alpha}
\alpha=\frac{a+\sqrt{a^{2}+4b}}{2} \qquad \text{and}\qquad \beta=\frac{a-\sqrt{a^{2}+4b}}{2}.
\end{equation}
\end{definition}
\begin{definition}
We define the $\mathcal{H}_{a,b}^{(s)}$ multiplicative arithmetic function as
\begin{equation*}
\mathcal{H}_{a,b}^{(s)}(p^{m})=H_{m}^{a,b}(1,s),
\end{equation*}
where $p$ is a prime number and $m$ is a non-negative integer.
\end{definition}

In \cite[Section 2]{SchMB}, the authors highlighted the relationship between the multiplicative function $\mathcal{F}_{a,b}$ and $(a,b)$-Fibonacci numbers, and on the other hand in \cite[Section 5]{Sch94} the relationship between the multiplicative function $\mathcal{L}_{a,b}$ and $(a,b)$-Lucas numbers.

\begin{theorem} \label{Th.first}
For complex numbers $a, b$ and integers $m, p$ ($m$ non-negative and $p$ prime) the following identities hold:
\begin{enumerate}
\item[(i)](\cite{SchMB, Sch94})\begin{equation*}
\mathcal{F}_{a,b}(p^{m})=F_{m+1}^{a,b} \qquad \text{and} \qquad \mathcal{L}_{a,b}(p^{m}) = \begin{cases}
1, & \text{\emph{if} $m=0$;} \\
L_{m}^{a,b}, & \text{\emph{if} $m>0$.}
\end{cases}
\end{equation*}
\item[(ii)]\begin{equation*}
\left[\frac{\mathcal{F}_{a,b}}{\mathcal{L}_{a,b}} \right]_{D}=\mathcal{F}_{0,-b} \qquad \text{and} \qquad \left[\frac{\mathcal{L}_{a,b}}{\mathcal{F}_{a,b}} \right]_{U}=\mathcal{H}_{a,b}^{(0)}=\mu_{a}\ast\mathcal{F}_{a,b},
\end{equation*}
where $\mu_{a}$ is the multiplicative arithmetic function defined by
\[ \mu_{a}(p^{m}) =
\begin{cases}
1, & \text{\emph{if} $m=0$;} \\
-a, & \text{\emph{if} $m=1$;} \\
0, & \text{\emph{if} $m>1$.}
\end{cases} \]
($\mu_{1}$ is the classical M\"{o}bius function, denoted by $\mu$ in the proof of Theorem \ref{Th.Szakacs}.)
\end{enumerate}
\end{theorem}
\begin{proof}
\leavevmode

(i) Since $\mathcal{F}_{a,b}$ and $\mathcal{L}_{a,b}$ are multiplicative, it is clear that
\begin{equation*}
\mathcal{F}_{a,b}(p^{0})=\mathcal{L}_{a,b}(p^{0})=1=F_{1}^{a,b}.
\end{equation*}
For $m>0$, using the Binet formulas (see (\ref{eq.Binet})) we get
\begin{equation*}
\mathcal{F}_{a,b}(p^{m})=(\alpha^{\Omega}\ast\beta^{\Omega})(p^{m})=\sum_{k=0}^{m}\alpha^{k}\beta^{m-k}=\frac{\alpha^{m+1}-\beta^{m+1}}{\alpha-\beta}=F_{m+1}^{a,b},
\end{equation*}
and
\begin{equation*}
\mathcal{L}_{a,b}(p^{m})=(\alpha^{\Omega}\sqcup\beta^{\Omega})(p^{m})=\alpha^{m}+\beta^{m}=L_{m}^{a,b}.
\end{equation*}

(ii)  It is straightforward to check that the inverse $\mathcal{F}_{a,b}^{-1}$ of $\mathcal{F}_{a,b}$ in the group $(\mathcal{M},\ast)$ is given by (see also \cite{SchMB})
\begin{equation} \label{eq.Fiboinverse}
\mathcal{F}_{a,b}^{-1}(p^{m}) = \begin{cases}
1,&\text{if $m=0$;} \\
-a, &\text{if $m=1$;} \\
-b, &\text{if $m=2$;} \\
0, &\text{if $m>2$.}
\end{cases}
\end{equation}
Obviously, if $m=0$ and $m=1$, then we have
\begin{equation*}
\mathcal{F}_{a,b}\ast\mathcal{F}_{0,-b}^{-1}(p^{m})=\mathcal{F}_{a,b}(p^{m})=\mathcal{L}_{a,b}(p^{m}).
\end{equation*}
Invoking Equation (\ref{eq.connect}) it follows that if $m>1$ then
\begin{equation*}
(\mathcal{F}_{a,b}\ast\mathcal{F}_{0,-b}^{-1})(p^{m})=\mathcal{F}_{a,b}(p^{m})+b\mathcal{F}_{a,b}(p^{m-2})=F_{m+1}^{a,b}+bF_{m-1}^{a,b}=L_{m}^{a,b}=\mathcal{L}_{a,b}(p^{m}).
\end{equation*}
We conclude that $\mathcal{F}_{a,b}=\mathcal{L}_{a,b}\ast\mathcal{F}_{0,-b}.$

Now, it is clear that $\mathcal{H}_{a,b}(1,0)(p^{0})=H_{0}^{a,b}(1,0)=1$, $\mathcal{H}_{a,b}(1,0)(p)=H_{1}^{a,b}(1,0)=0$, and by induction we get
\begin{equation} \label{H(1,0)}
\mathcal{H}_{a,b}(1,0)(p^{m})=H_{m}^{a,b}(1,0) = \begin{cases}
1,&\text{if $m=0$;} \\
bF_{m-1}^{a,b}, & \text{if $m>0$.}
\end{cases}
\end{equation}
while
\begin{equation*}
H_{m+1}^{a,b}(1,0)=aH_{m}^{a,b}(1,0)+bH_{m-1}^{a,b}(1,0)=abF_{m-1}^{a,b}+b^{2}F_{m-2}^{a,b}=bF_{m}^{a,b}
\end{equation*}
if $m>1$. Invoking Equations (\ref{H(1,0)}) and (\ref{eq.connect}), we may conclud that
\begin{equation*}
(\mathcal{F}_{a,b}\sqcup\mathcal{H}_{a,b}(1,0))(p^{m})=F_{m+1}^{a,b}+bF_{m-1}^{a,b}=L_{m}^{a,b} \qquad (m>0),
\end{equation*}
and therefore $\mathcal{L}_{a,b}=\mathcal{F}_{a,b}\sqcup\mathcal{H}_{a,b}(1,0).$

Finally, since $(\mu_{a}\ast\mathcal{F}_{a,b})(p^{0})=1$ and
\begin{equation*}
(\mu_{a}\ast\mathcal{F}_{a,b})(p^{m})=\sum_{i=0}^{m}\mu_{a}(p^{i})\mathcal{F}_{a,b}(p^{m-i})=F_{m+1}^{a,b}-aF_{m}^{a,b}=bF_{m-1}^{a,b}
\end{equation*}
if $m>0$, the proof is complete.
\end{proof}
Also taking into account the quasi-distributive law (\ref{eq.Distributive}), we arrive at the next well-known Fibonacci-Lucas identity.
\begin{corollary} \label{Remark}
For $(a,b)$-Fibonacci and $(a,b)$-Lucas numbers, and $\Delta=a^{2}+4b\neq0$, the following identity holds:
\begin{equation}
\Delta{F_{m+1}^{a,b}}=L_{m+2}^{a,b}+bL_{m}^{a,b}.  \label{Delta}
\end{equation}
\end{corollary}
\begin{proof}
Since $\mathcal{F}_{0,-b}^{-1}\ast\mathcal{L}_{a,b}=\mathcal{F}_{0,-b}^{-1}\ast(\mathcal{F}_{a,b}\sqcup\mathcal{H}_{a,b}^{0})$, and $\mathcal{F}_{0,-b}^{-1}(p^{m})=0$ if $m>2$, by the quasi-distributive law (\ref{eq.Distributive}) we have
\begin{equation*}
(\mathcal{F}_{0,-b}^{-1}\ast\mathcal{L}_{a,b})(p^{m+2})=(\mathcal{F}_{0,-b}^{-1}\ast\mathcal{F}_{a,b})(p^{m+2})+(\mathcal{F}_{0,-b}^{-1}\ast\mathcal{H}_{a,b}^{0})(p^{m+2}) \quad \text{if}\ m>0.
\end{equation*}
Thus,
\begin{align*}
L_{m+2}^{a,b}+bL_{m}^{a,b} & =F_{m+3}^{a,b}+bF_{m+1}^{a,b}+bF_{m+1}^{a,b}+b^{2}F_{m-1}^{a,b} \\
 & =aF_{m+2}^{a,b}+4bF_{m+1}^{a,b}-abF_{m}^{a,b} \\
 & =a^{2}F_{m+1}^{a,b}+4bF_{m+1}^{a,b}=\Delta{F_{m+1}^{a,b}}.
\end{align*}
The proof is complete.
\end{proof}
In the special case of $k$-Fibonacci numbers we have
\begin{equation} \label{eq.k-Fibonacci}
(k^{2}+4)F_{k,m+1}=L_{k,m+2}+L_{k,m}.
\end{equation}



\section{Applications to Fibonacci-Lucas relations} \label{sect.:Appl.}

\subsection{Sums involving consecutive even and odd subscript $(a,b)$-Lucas numbers} \label{Subs.3.1}
As an initial application, in Theorem \ref{Th. Lucas par-impar} (a) we use the convolution tool of arithmetic functions to prove two sums that involve the first $n$ consecutive odd and even subscript $(a,b)$-Lucas numbers. Then we extend these sums in part (b).
\begin{theorem} \label{Th. Lucas par-impar}
For any non-negative integers $m$ and $n,$ and $b\neq0$ we have the following:
\begin{enumerate}
\item[(a)]
\begin{enumerate}
\item[(i)]
$\sum_{i=0}^{n}(-b)^{n-i}L_{2i+1}^{a,b}=F_{2n+2}^{a,b};$
\item[(ii)]
$\sum_{i=0}^{n}(-b)^{n-i}L_{2i}^{a,b}=F_{2n+1}^{a,b}+(-b)^{n};$
\end{enumerate}
\item[(b)]
\begin{enumerate}
\item[(i)]\begin{align*}
\sum_{i=0}^{n}(-b)^{n-i}L_{2(i+m)+1}^{a,b} & =F_{2(n+m)+2}^{a,b}+H_{2m+1}^{a,b}((-b)^{n},0) \\
& =F_{2(n+m)+2}^{a,b}+(-1)^{n}b^{n+1}F_{2m}^{a,b} \\
& =F_{n+m+1}^{a,b}L_{n+m+1}^{a,b}+(-1)^{n}b^{n+1}F_{m}^{a,b}L_{m}^{a,b};
\end{align*}
\item[(ii)]\begin{align*}
\sum_{i=0}^{n}(-b)^{n-i}L_{2(i+m)}^{a,b} & =F_{2(n+m)+1}^{a,b}+H_{2m}^{a,b}((-b)^{n},0)\\
& =F_{2(n+m)+1}^{a,b}+(-1)^{n}b^{n+1}F_{2m-1}^{a,b}\\
& =F_{n+m+1}^{a,b}L_{n+m}^{a,b}+(-1)^{n}b^{n+1}F_{m}^{a,b}L_{m-1}^{a,b}.
\end{align*}
\end{enumerate}
\end{enumerate}
\end{theorem}
\begin{proof}
\leavevmode

(a) We use the first quotient $\left[\frac{\mathcal{F}_{a,b}}{\mathcal{L}_{a,b}} \right]_{D}=\mathcal{F}_{0,-b}$, that is $\mathcal{F}_{a,b}=\mathcal{L}_{a,b}\ast\mathcal{F}_{0,-b}$. It is straightforward to check that $\mathcal{F}_{0,-b}$ is given by
\begin{equation*}
\mathcal{F}_{0,-b}(p^{m}) = \begin{cases}
(-b)^{\frac{m}{2}}, & \text{if \emph{m} is an even integer;} \\
0, & \text{if \emph{m} is an odd integer.}
\end{cases}
\end{equation*}

(i) Let $\ell=2n+1$. Then
\begin{equation*}
F_{2n+2}^{a,b}=\mathcal{F}_{a,b}(p^{\ell})=\sum_{i=0}^{\ell}\mathcal{L}_{a,b}(p^{i})\mathcal{F}_{0,-b}(p^{\ell-i})=\sum_{i=0}^{n}(-b)^{n-i}L_{2i+1}^{a,b}.
\end{equation*}

(ii) If $\ell=2n$ then
\begin{align*}
F_{2n+1}^{a,b}+(-b)^{n} & =\mathcal{F}_{a,b}(p^{\ell})+(-b)^{n} \\
& =\sum_{i=0}^{\ell}\mathcal{L}_{a,b}(p^{i})\mathcal{F}_{0,-b}(p^{\ell-i})+(-b)^{n} \\
& =\mathcal{F}_{0,-b}(p^{\ell})+\sum_{i=1}^{n}L_{2i}^{a,b}\mathcal{F}_{0,-b}(p^{2n-2i})+(-b)^{n} \\
& =\sum_{i=0}^{n}(-b)^{n-i}L_{2i}^{a,b}.
\end{align*}

(b) We prove the first part of $b(i)$ and $b(ii)$ by induction on $m.$ For $m=0 $ see (a). For $m=1$ we get
\begin{align*}
\sum_{i=0}^{n}(-b)^{n-i}L_{2(i+1)+1}^{a,b} & =\sum_{i=0}^{n}(-b)^{n-i}(aL_{2(i+1)}^{a,b}+bL_{2i+1}^{a,b}) \\
& =aF_{2n+3}^{a,b}+aH_{2}^{a,b}((-b)^{n},0)+bF_{2n+2}^{a,b} \\
& =F_{2n+4}^{a,b}+H_{3}^{a,b}((-b)^{n},0).
\end{align*}
and
\begin{align*}
\sum_{i=0}^{n}(-b)^{n-i}L_{2(i+1)}^{a,b} & =\sum_{i=0}^{n}(-b)^{n-i}(aL_{2i+1}^{a,b}+bL_{2i}^{a,b}) \\
& =aF_{2n+2}^{a,b}+bF_{2n+1}^{a,b}+(-1)^{n}b^{n+1} \\
& =F_{2n+3}^{a,b}+H_{2}^{a,b}((-b)^{n},0),
\end{align*}
Thus we can proceed to the second step of induction ($m>1$). We have
\begin{align*}
 \sum_{i=0}^{n}(-b)^{n-i}L_{2(i+m)+1}^{a,b} & =\sum_{i=0}^{n}(-b)^{n-i}(aL_{2(i+m)}^{a,b}+bL_{2(i+m)-1}^{a,b}) \\
& =aF_{2(n+m)+1}^{a,b}+aH_{2m}^{a,b}((-b)^{n},0)+bF_{2(n+m)}^{a,b}+bH_{2m-1}^{a,b}((-b)^{n},0) \\
& =F_{2(n+m)+2}^{a,b}+H_{2m+1}^{a,b}((-b)^{n},0).
\end{align*}
and
 \begin{align*}
 \sum_{i=0}^{n}(-b)^{n-i}L_{2(i+m)}^{a,b} & =\sum_{i=0}^{n}(-b)^{n-i}(aL_{2(i+m)-1}^{a,b}+bL_{2(i+m)-2}^{a,b}) \\
& =aF_{2(n+m)}^{a,b}+aH_{2m-1}^{a,b}((-b)^{n},0)+bF_{2(n+m)-1}^{a,b}+bH_{2m-2}^{a,b}((-b)^{n},0) \\
& =F_{2(n+m)+1}^{a,b}+H_{2m}^{a,b}((-b)^{n},0),
\end{align*}
The proof of the first part of (b) (i) and (b) (ii) is complete. By applying the following lemma, the second part of our proof is also solved.
\begin{lemma}
For all non-negative integer $m$, the following identity holds:
\begin{equation*}
H_{m}^{a,b}((-b)^{n},0)=(-1)^{n}b^{n+1}F_{m-1}^{a,b}.
\end{equation*}
\end{lemma}
The proof of this lemma is by induction on $m$ and is omitted.

Now, before proceeding to the proof of the last part we make the following remark.
\begin{remark}
Bitim and Topal \cite[Corollary 2]{Bitim} established the following Fibonacci-Lucas relation:
\begin{equation*}
\sum_{i=0}^{n}(-b)^{i}L_{2i}^{a,b}=(-b)^{-n}F_{n+1}^{a,b}L_{n}^{a,b}.
\end{equation*}
This relation together with Part (a) (ii) of Theorem \ref{Th. Lucas par-impar}, gives the following formula for odd subscript $(a,b)$-Fibonacci numbers:
\begin{equation} \label{eq.Fibimpar}
F_{2n+1}^{a,b}=F_{n+1}^{a,b}L_{n}^{a,b}+(-1)^{n+1}b^{n}.
\end{equation}
The identity (\ref{eq.Fibimpar}) is missing from the fairly comprehensive list of connected formulas (2.1)-(2.13) inserted in \cite{Bitim}, but the twin formula (with even subscript),
\begin{equation} \label{eq.Fibpar}
F_{2n}^{a,b}=F_{n}^{a,b}L_{n}^{a,b},
\end{equation}
appears in \cite[(2.3)]{Bitim}.
\end{remark}
Now, using these two Fibonacci-Lucas identities (\ref{eq.Fibimpar}) and (\ref{eq.Fibpar}), the last part of the theorem becomes an immediate consequence.
\end{proof}
\begin{remark}
Perhaps the simplest proof of Theorem \ref{Th. Lucas par-impar}(a) is the one below, using (\ref{eq.connect}) and then rearranging the parentheses.
\begin{align*}
\sum_{i=0}^{n}(-b)^{n-i}L_{2i+1}^{a,b} & =\sum_{i=0}^{n}(-b)^{n-i}(F_{2i+2}^{a,b}+bF_{2i}^{a,b}) \\
& =(-b)^{n}bF_{0}^{a,b}+\sum_{i=0}^{n-1}((-b)^{n-i}F_{2i+2}^{a,b}+(-b)^{n-i-1}bF_{2i+2}^{a,b})+(-b)^{0}F_{2n+2}^{a,b}=F_{2n+2}^{a,b}.
\end{align*}
This procedure proves the Jacobsthal version of Theorem \ref{Th. Lucas par-impar}(a) in \cite{SchAMM}. The perspective offered by the initial proof in Theorem \ref{Th. Lucas par-impar}(a) is broader and highlights the interdependence of the two chapters of number theory: multiplicative arithmetic functions and Fibonacci numbers (in this regard, see also \cite[Section 2]{SchMB}).
\end{remark}

\begin{corollary}
The $k$-Fibonacci version of Theorem \ref{Th. Lucas par-impar}(b)(ii) is the following:
\begin{equation*}
\sum_{i=0}^{n}(-1)^{i}L_{k,2(i+m)}=(-1)^{n}F_{k,2(n+m)+1}+F_{k,2m-1}.
\end{equation*}
\end{corollary}
This identity is fully consistent with the following formula given in \cite[Theorem 3.4]{Panwar}:
\begin{equation*}
\sum_{i=0}^{n}(-1)^{i}L_{k,2(i+m)}=\frac{1}{k^{2}+4}(L_{k,2m-2}-L_{k,2(n+m)}+L_{k,2m}-L_{k,2(n+m+1)}),
\end{equation*}
only if $n$ is odd, since $L_{k,u}+L_{k,u+2}=(k^{2}+4)F_{k,u+1}$ (see Equation (\ref{eq.k-Fibonacci})). If the non-negative integer $n$ is even then the two minus signs before $L_{k,2(n+m)}$ and $L_{k,2(n+m+1)}$ should have been plus.

\subsection{Regarding even and odd subscript $(a,b)$-Fibonacci numbers} \label{Subs.3.2}
In this subsection, the $(a,b)$-Lucas numbers from Theorem \ref{Th. Lucas par-impar} (a) are replaced by the $(a,b)$-Fibonacci numbers; and the new Fibonacci-Lucas relations are proven directly using Equation (\ref{Delta}). Taking into consideration Corollary \ref{Remark}, we can observe that the origin of these relations is the Fibonacci-Lucas quotients (ii) of Theorem \ref{Th.first} and the quasi-distributive law (\ref{eq.Distributive}). Then we consider particular cases referring to ordinary Fibonacci, Pell and Jacobsthal numbers.
\begin{theorem} \label{Th.Fib.}
For all non-negative integers $n$, and $\Delta=a^{2}+4b\neq0$, we have
\begin{enumerate}
\item[(a)]
\begin{enumerate}
\item[(i)]  $\sum_{i=0}^{n}(-b)^{n-i}F_{2i}^{a,b}=\frac{1}{\Delta}(L_{2n+1}^{a,b}+(-1)^{n+1}ab^{n});$
\item[(ii)] $\sum_{i=0}^{n}(-b)^{n-i}F_{2i+1}^{a,b}=\frac{1}{\Delta}(L_{2n+2}^{a,b}+(-1)^{n}2b^{n+1});$
\end{enumerate}
\item[(b)]
\begin{enumerate}
\item[(i)]  $\sum_{i=0}^{n}(-1)^{i}F_{2i}=\frac{1}{5}((-1)^{n}L_{2n+1}-1); \qquad \sum_{i=0}^{n}(-1)^{i}L_{2i}=(-1)^{n}F_{2n+1}+1;$
\item[(ii)] $\sum_{i=0}^{n}(-1)^{i}F_{2i+1}=\frac{1}{5}((-1)^{n}L_{2n+2}+2); \quad \sum_{i=0}^{n}(-1)^{i}L_{2i+1}=(-1)^{n}F_{2n+2};$
\end{enumerate}
\item[(c)]
\begin{enumerate}
\item[(i)]  $\sum_{i=0}^{n}(-1)^{i}P_{2i}=\frac{1}{8}((-1)^{n}Q_{2n+1}-2); \qquad \sum_{i=0}^{n}(-1)^{i}Q_{2i}=(-1)^{n}P_{2n+1}+1;$
\item[(ii)] $\sum_{i=0}^{n}(-1)^{i}P_{2i+1}=\frac{1}{8}((-1)^{n}Q_{2n+2}+2); \quad \sum_{i=0}^{n}(-1)^{i}Q_{2i+1}=(-1)^{n}P_{2n+2};$
\end{enumerate}
\item[(d)]
\begin{enumerate}
\item[(i)]  $\sum_{i=0}^{n}(-1)^{i}2^{n-i}J_{2i}=\frac{1}{9}((-1)^{n}j_{2n+1}-2^{n});$\\ $\sum_{i=0}^{n}(-1)^{i}2^{n-i}j_{2i}=(-1)^{n}J_{2n+1}+2^{n};$
\item[(ii)] $\sum_{i=0}^{n}(-1)^{i}2^{n-i}J_{2i+1}=\frac{1}{9}((-1)^{n}j_{2n+2}+2^{n+2});$\\ $\sum_{i=0}^{n}(-1)^{i}2^{n-i}j_{2i+1}=(-1)^{n}J_{2n+2};$\\
(for the alternating Jacobsthal-Lucas sums see \cite{SchAMM}).
\end{enumerate}
\end{enumerate}
\end{theorem}
\begin{proof}
\leavevmode

(a) Using the formula $\Delta{F_{n}^{a,b}=L_{n+1}^{a,b}+bL_{n-1}^{a,b}}$ (see Equation (\ref{Delta})), where $\Delta=a^{2}+4b$, we get

(i) \begin{align*}
\sum_{i=0}^{n}(-b)^{n-i}L_{2i+1}^{a,b} & =\Delta\sum_{i=0}^{n}(-b)^{n-i}F_{2i}^{a,b}-b\sum_{i=0}^{n}(-b)^{n-i}L_{2i-1}^{a,b} \\
& =\Delta\sum_{i=0}^{n}(-b)^{n-i}F_{2i}^{a,b}+(-b)^{n+1}(-\frac{a}{b})+\sum_{i=0}^{n}(-b)^{n-i}L_{2i+1}^{a,b}-L_{2n+1}^{a,b}.
\end{align*}
Hence
\begin{equation*}
\Delta\sum_{i=0}^{n}(-b)^{n-i}F_{2i}^{a,b}=L_{2n+1}^{a,b}+(-1)^{n+1}ab^{n},
\end{equation*}
which is the desired identity.

\medskip

(ii)
\begin{align*}
\sum_{i=0}^{n}(-b)^{n-i}L_{2i}^{a,b} & =\frac{\Delta}{b}\sum_{i=0}^{n}(-b)^{n-i}F_{2i+1}^{a,b}-\frac{1}{b}\sum_{i=0}^{n}(-b)^{n-i}L_{2i+2}^{a,b} \\
& =\frac{\Delta}{b}\sum_{i=0}^{n}(-b)^{n-i}F_{2i+1}^{a,b}+\sum_{i=0}^{n}(-b)^{n-i}L_{2i}^{a,b}+(-1)^{n+1}2b^{n}-\frac{1}{b}L_{2n+2}^{a,b} \\
& =\frac{\Delta}{b}\sum_{i=0}^{n}(-b)^{n-i}F_{2i+1}^{a,b}-(-b)^{n}L_{0}^{a,b}+\sum_{i=0}^{n}(-b)^{n-i}L_{2i}^{a,b}-\frac{1}{b}L_{2n+2}^{a,b}.
\end{align*}
Hence
\begin{equation*}
\frac{\Delta}{b}\sum_{i=0}^{n}(-b)^{n-i}F_{2i+1}^{a,b}=\frac{1}{b}L_{2n+2}^{a,b}+(-b)^{n}2.
\end{equation*}
The proof of (a) is now complete.

The formulas (b), (c), and (d) are special cases of Theorems \ref{Th.Fib.}($a$) and \ref{Th. Lucas par-impar}($a$) for the Fibonacci (Lucas), Pell (Pell-Lucas), and Jacobsthal (Jacobsthal-Lucas) numbers, respectively.
\end{proof}
The established formulas are of Fibonacci-Lucas type in the sense that both numbers (Fibonacci and Lucas) are present in the formula. Of course, we can find more friendly formulas
 by giving up this restriction . For instance, by eliminating the $(a,b)$-Lucas number $L_{2n+1}^{a,b}$ from Theorem \ref{Th.Fib.} (a)(i), we get the next corollary.
\begin{corollary}
For $(a,b)$-Fibonacci numbers the following identity holds:
\begin{equation*}
\sum_{i=0}^{n}(-b)^{n-i}F_{2i}^{a,b}=F_{n}^{a,b}F_{n+1}^{a,b}.
\end{equation*}
\end{corollary}
\begin{proof}
Using Equations (\ref{eq.connect}), (\ref{Delta}), (\ref{eq.Fibimpar}), and (\ref{eq.Fibpar}) we get
\begin{align*}
\Delta{F_{n}^{a,b}}F_{n+1}^{a,b} & =(L_{n+1}^{a,b}+bL_{n-1}^{a,b})F_{n+1}^{a,b} \\ & =F_{2n+2}^{a,b}+(L_{n+1}^{a,b}-aL_{n}^{a,b})F_{n+1}^{a,b} \\
& = F_{2n+2}^{a,b}+F_{2n+2}^{a,b}-aL_{n}^{a,b}F_{n+1}^{a,b} \\ &= F_{2n+2}^{a,b}+F_{2n+2}^{a,b}-a(F_{2n+1}^{a,b}+(-b)^{n}) \\
& = F_{2n+2}^{a,b}+bF_{2n}^{a,b}+(-1)^{n+1}ab^{n} \\ & =L_{2n+1}^{a,b}+(-1)^{n+1}ab^{n}.
\end{align*}
The proof is complete by Theorem \ref{Th.Fib.} (a)(i).
\end{proof}

\subsection{Convolution sums involving $(a,b)$-Fibonacci and $(a,b)$-Lucas numbers}
As a new application of our approach, we
start with a proof of a theorem using the Dirichlet inverse of the multiplicative arithmetic function $\mathcal{F}_{a,b}$. Then Kesava Menon quasi-distributive law  leads to the Fibonacci-Lucas convolution sum. A specific case reveals the identities (\ref{eq.Griffiths}) and (\ref{eq.Koshy}) mentioned in Section~\ref{sec1}.
\begin{theorem}
Let $c$ be a non-zero complex number such that $c^{2}-ac-b\neq0$ and let $n$ be a positive integer. We have
\begin{equation}\ \label{eq.fallingpower}
\sum_{i=1}^{n}c^{n-i}F_{i}^{a,b}=\frac{c^{n+1}-cF_{n+1}^{a,b}-bF_{n}^{a,b}}{c^{2}-ac-b}
\end{equation}
\end{theorem}
\begin{proof}
If $f=\mathcal{F}_{a,b}\ast{c}^{\Omega}$ then $f(p^{n-1})=\sum_{i=0}^{n-1}\mathcal{F}_{a,b}(p^{i})c^{\Omega(p^{n-i})}=\sum_{i=1}^{n}c^{n-i}F_{i}^{a,b},$ and $c^{\Omega}=\mathcal{F}_{a,b}^{-1}\ast{f}.$

Now, Equation (\ref{eq.fallingpower}) holds for $n=1.$ We assume that $n>1$. Invoking Equation (\ref{eq.Fiboinverse}) we have
\begin{align*}
c^{n} & =c^{\Omega(p^{n})}=(\mathcal{F}_{a,b}^{-1}\ast{f})(p^{n})=\sum_{i=1}^{n+1}c^{n+1-i}F_{i}^{a,b}-a\sum_{i=1}^{n}c^{n-i}F_{i}^{a,b}-b\sum_{i=1}^{n-1}c^{n-1-i}F_{i}^{a,b}\\
& =c\sum_{i=1}^{n}c^{n-i}F_{i}^{a,b}+F_{n+1}^{a,b}-a\sum_{i=1}^{n}c^{n-i}F_{i}^{a,b}-\frac{b}{c}\sum_{i=1}^{n}c^{n-i}F_{i}^{a,b}+\frac{b}{c}F_{n}^{a,b},
\end{align*}
and the proof is finished.
\end{proof}
In what follows, a fundamental role is played by the Kesava Menon quasi-distributive law (\ref{eq.Distributive}).
\begin{theorem} \label{Th.compl}
We have
\begin{equation} \label{eq.complicata}
\sum_{i=1}^{n}L_{n-i}^{a,b}F_{i}^{c,d}=\frac{\alpha^{n+1}-\alpha{F}_{n+1}^{c,d}-dF_{n}^{c,d}}{\alpha^{2}-c\alpha-d}+\frac{\beta^{n+1}-\beta{F}_{n+1}^{c,d}-dF_{n}^{c,d}}{\beta^{2}-c\beta-d},
\end{equation}
if any of the roots $\alpha,\beta$ of the quadratic equation $x^{2}-ax-b=0$ (see (\ref{eq.alpha})) are not zeros of $x^{2}-cx-d$.
\end{theorem}
\begin{proof}
Using the quasi-distributive law (\ref{eq.Distributive}) we get
\begin{equation*}
(\mathcal{F}_{c,d}\ast\mathcal{L}_{a,b})\sqcup\mathcal{F}_{c,d}=(\mathcal{F}_{c,d}\ast\alpha^{\Omega})\sqcup(\mathcal{F}_{c,d}\ast\beta^{\Omega}).
\end{equation*}
Since $(\mathcal{F}_{c,d}\ast\mathcal{L}_{a,b})(p^{n-1})=\sum_{i=1}^{n}L_{n-i}^{a,b}F_{i}^{c,d}-F_{n}^{c,d}$ we have
\begin{equation*}
[(\mathcal{F}_{c,d}\ast\mathcal{L}_{a,b})\sqcup\mathcal{F}_{c,d}](p^{n-1})=\sum_{i=1}^{n}L_{n-i}^{a,b}F_{i}^{c,d}.
\end{equation*}
Invoking Equation (\ref{eq.fallingpower}), it follows that
\begin{align*}
[(\mathcal{F}_{c,d}\ast\alpha^{\Omega})\sqcup(\mathcal{F}_{c,d}\ast\beta^{\Omega})](p^{n-1}) & =\sum_{i=1}^{n}\alpha^{n-i}F_{i}^{c,d}+\sum_{i=1}^{n}\beta^{n-i}F_{i}^{c,d} \\
& =\frac{\alpha^{n+1}-\alpha{F}_{n+1}^{c,d}-dF_{n}^{c,d}}{\alpha^{2}-c\alpha-d}+\frac{\beta^{n+1}-\beta{F}_{n+1}^{c,d}-dF_{n}^{c,d}}{\beta^{2}-c\beta-d},
\end{align*}
and the proof is finished.
\end{proof}
In the case of $a=c$ and $b\neq{d}$, Equation (\ref{eq.complicata}) is significantly simplified.
We thus arrive (via the Kesava Menon quasi-distributive law) to the Equations (\ref{eq.Griffiths}) and (\ref{eq.Koshy}) mentioned at the beginning of the paper.
\begin{corollary} \label{Cor.}
If $b\neq{d}$, then
\begin{enumerate}
\item[(a)] \begin{equation} \label{eq.mixt}  \sum_{i=1}^{n}L_{n-i}^{a,b}F_{i}^{a,d}=\frac{L_{n+1}^{a,b}-L_{n+1}^{a,d}}{b-d}=\sum_{i=1}^{n}L_{n-i}^{a,d}F_{i}^{a,b};
\end{equation}
\item[(b)]
\begin{enumerate}
\item[(i)] $(\cite[\emph{Section}\ 2]{Griffiths}$, $\cite[\emph{Theorem}\ 1.1]{Bramham}$ and $\cite[(2.6),(2.7)]{Koshy}$---see also (\ref{eq.Griffiths}) and (\ref{eq.Koshy}))
\begin{equation*}
\sum_{i=1}^{n}L_{n-i}J_{i}=j_{n+1}-L_{n+1}=\sum_{i=1}^{n}j_{n-i}F_{i};
\end{equation*}
\item[(ii)]
\begin{equation*}
2\sum_{i=1}^{n}P_{i}=\frac{1}{2}(Q_{n+1}-2)=\sum_{i=1}^{n}iQ_{n-i};
\end{equation*}
\item[(iii)]
\begin{equation*}
\sum_{i=1}^{n}L_{2(n-i)}(2^{i}-1)=L_{2n+2}-2^{n+1}-1=\sum_{i=1}^{n}(2^{n-i}+1)F_{2i}.
\end{equation*}
\end{enumerate}
\end{enumerate}
\end{corollary}
\begin{proof}
(a) Using Equation (\ref{eq.complicata}), we get
\begin{align*}
\sum_{i=1}^{n}L_{n-i}^{a,b}F_{i}^{a,d} & =\frac{\alpha^{n+1}-\alpha{F}_{n+1}^{a,d}-dF_{n}^{a,d}}{\alpha^{2}-a\alpha-d}+\frac{\beta^{n+1}-\beta{F}_{n+1}^{a,d}-dF_{n}^{a,d}}{\beta^{2}-a\beta-d} \\
& =\frac{\alpha^{n+1}+\beta^{n+1}-aF_{n+1}^{a,d}-2dF_{n}^{a,d}}{b-d}=\frac{L_{n+1}^{a,b}-L_{n+1}^{a,d}}{b-d} \\
& =\frac{L_{n+1}^{a,d}-L_{n+1}^{a,b}}{d-b}=\sum_{i=1}^{n}L_{n-i}^{a,d}F_{i}^{a,b}.
\end{align*}

(b) It is straightforward to see that (i), (ii), and (iii) follow immediately from (a).

\medskip

(i) The ordinary Fibonacci and Lucas sequences $\{F_{n}\}_{n\geq0}$ and $\{L_{n}\}_{n\geq0}$, respectively, are recovered when $a=b=1$, and the Jacobsthal and Jacobsthal-Lucas sequences $\{J_{n}\}_{n\geq0}$ and $\{j_{n}\}_{n\geq0}$, respectively, are recovered when $a=1$ and $b=2$.

\medskip

(ii) The sequence of non-negative integers and the constant sequence $2,2,2,\ldots$ are the $(2,-1)$-Fibonacci and the $(2,-1)$-Lucas sequences, respectively. The Pell and Pell-Lucas sequences are the
$(2,1)$-Fibonacci and the $(2,1)$-Lucas sequences, respectively.

\medskip

(iii). The sequences $(2^{n}-1)$ and $(2^{n}+1)$ are the Mersenne sequence (which is the $(3,-2)$-Fibonacci sequence) and the Mersenne-Lucas sequence (the $(3,-2)$-Lucas sequence), respectively. The even-numbered Fibonacci and Lucas sequences are the $(3,-1)$-Fibonacci and the $(3,-1)$-Lucas sequences, respectively.
\end{proof}


\subsection{The Kesava Menon quasi-distributive law again} \label{Subs.3.4}
 As a final application to our approach, we prove the $(a,b)$-Fibonacci convolution sum using again the quasi-distributive law (\ref{eq.Distributive}).
\begin{theorem} $(\cite[Theorem\ 4]{Szakacs})$ \label{Th.Szakacs}
For $(a,b)$-Fibonacci and $(a,b)$-Lucas numbers, and $\Delta=a^{2}+4b\neq0$, we have
\begin{equation} \label{eq.Th}
\Delta\sum_{i=0}^{n}F_{i}^{a,b}F_{n-i}^{a,b}=(n+1)L_{n}^{a,b}-2F_{n+1}^{a,b}.
\end{equation}
\end{theorem}
\begin{proof}
Let $\widetilde{\mathcal{F}}_{a,b}$ be the multiplicative arithmetic function defined by
\begin{equation*}
\widetilde{\mathcal{F}}_{a,b}=(\alpha^{\Omega}\ast\beta^{\Omega}\mu)\sqcup(\beta^{\Omega}\ast\alpha^{\Omega}\mu).
\end{equation*}
Note that if $f$ is a completely multiplicative arithmetic function then the usual product $f\mu$ is the inverse element of $f$ in the commutative group $(\mathcal{M},\ast)$. Thus, using the quasi-distributive law (\ref{eq.Distributive}) we get
\begin{equation} \label{eq.I}
\mathcal{F}_{a,b}\ast\widetilde{\mathcal{F}}_{a,b}\sqcup\mathcal{F}_{a,b}=(\alpha^{\Omega}\ast\alpha^{\Omega})\sqcup(\beta^{\Omega}\ast\beta^{\Omega}).
\end{equation}
We observe that if $p$ is a prime number then $\widetilde{\mathcal{F}}_{a,b}(p)=0$, and if $n>1$ then
\begin{equation*}
\widetilde{\mathcal{F}}_{a,b}(p^{n})=\alpha^{n}-\alpha^{n-1}\beta+\beta^{n}-\beta^{n-1}\alpha=\alpha^{n}+\beta^{n}-\alpha\beta(\alpha^{n-2}+\beta^{n-2})=L_{n}^{a,b}+bL_{n-2}^{a,b}.
\end{equation*}
So, by Equation (\ref{Delta}), we have
\begin{equation*}
\widetilde{\mathcal{F}}_{a,b}(p^{n}) = \begin{cases}
1,&\text{if $n=0$;} \\
\Delta{F}_{n-1}^{a,b}, & \text{if $n>0$.}
\end{cases}
\end{equation*}
Now, since
\begin{align*}
[(\mathcal{F}_{a,b}\ast\widetilde{\mathcal{F}}_{a,b})\sqcup\mathcal{F}_{a,b}](p^{n}) & =\sum_{i=0}^{n}\mathcal{F}_{a,b}(p^{n-i})\widetilde{\mathcal{F}}_{a,b}(p^{i})+\mathcal{F}_{a,b}(p^{n}) \\
& =F_{n+1}^{a,b}+\Delta\sum_{i=0}^{n}F_{i}^{a,b}F_{n-i}^{a,b}+F_{n+1}^{a,b},
\end{align*}
and
\begin{equation*}
[(\alpha^{\Omega}\ast\alpha^{\Omega})\sqcup(\beta^{\Omega}\ast\beta^{\Omega})](p^{n})=(n+1)\alpha^{n}+(n+1)\beta^{n}=(n+1)L_{n}^{a,b},
\end{equation*}
the proof is complete by (\ref{eq.I}).
\end{proof}

\section{Acknowledgment}
The author would like to thank the referee for carefully reading the paper and for providing many useful comments and suggestions.




\bigskip
\hrule
\bigskip

\noindent 2020 {\it Mathematics Subject Classification}: Primary 11B39; Secondary 11A25.

\noindent\emph{Keywords:} $(a,b)$-Fibonacci sequence, $(a,b)$-Lucas sequence, Fibonacci-Lucas relation, multiplicative arithmetic function, Dirichlet convolution, unitary convolution.

\bigskip
\hrule
\bigskip

\noindent (Concerned with sequences 
\seqnum{A000032},
\seqnum{A000045},
\seqnum{A000104},
\seqnum{A000129},
\seqnum{A001045},
\seqnum{A002203}, and
\seqnum{A014551}.)

\bigskip
\hrule
\bigskip

\vspace*{+.1in}
\noindent
Received July 31 2024; 
revised versions received August 19 2024; January 16 2025; January 22 2025; February 13 2025.
Published in {\it Journal of Integer Sequences}, February 13 2025.

\bigskip
\hrule
\bigskip

\noindent
Return to
\htmladdnormallink{Journal of Integer Sequences home page}{https://cs.uwaterloo.ca/journals/JIS/}.
\vskip .1in

\end{document}

